Author’s response to reviews

Title: Stereotactic radiotherapy on brain metastases with recent hemorrhagic signal: STEREO-HBM, a two-step phase 2 trial

Authors:
Paul LESUEUR (p.lesueur@baclesse.unicancer.fr)
William KAO (w.kao@baclesse.unicancer.fr)
Alexandra LECONTE (a.leconte@baclesse.unicancer.fr)
Julien GEFFRELOT (j.geffrelot@baclesse.unicancer.fr)
Justine LEQUESNE (j.lequesne@baclesse.unicancer.fr)
Joelle LACROIX (j.lacroix@baclesse.unicancer.fr)
Pierre-Emmanuel BRACHET (pe.brachet@baclesse.unicancer.fr)
Iona HRAB (i.hrab@baclesse.unicancer.fr)
Philippe Royer (p.royer@nancy.unicancer.fr)
Benedicte CLARISSE (b.clarisse@baclesse.unicancer.fr)
Dinu STEFAN (d.stefan@baclesse.unicancer.fr; b.clarisse@baclesse.fr)

Version: 1 Date: 17 Jan 2020

Author’s response to reviews:

Caen, January 17, 2020

BMC Cancer
Editor-in-Chief

Ref: BCAN-D-19-03783

Dear Editor,

We gratefully acknowledge you for considering our submission to BMC Cancer related to the study protocol entitled "Stereotactic radiotherapy on brain metastases with recent hemorrhagic signal: STEREO-HBM, a two-step phase 2 trial".
We have carefully considered your minor comments and have revised the file:

1. To distinguish between the two authors who have the same initials (JL), we have designated them as JL1 and JL2, with JL1 corresponding to Justine Lequese, in the Authors’ Contributions section.

2. the Authors’ Contributions section has been revised to precise the contribution of Iona Harb (IH)

We hope these precisions will meet the requirements of your valuable journal and may be considered for publication.

Best regards,

Bénédicte CLARISSE

Clinical Research Department
b.clarisse@baclesse.fr

Centre François Baclesse, 3 avenue Général Harris, 14076 Caen Cedex 5, France